Author's response to reviews

Title: GRAIDs: A framework for closing the gap in the availability of health promotion programs and interventions for people with disabilities

Authors:

James H Rimmer (jrimmer@uab.edu)
Kerri A Vanderbom (kerriv@lakeshore.org)
Linda Bandini (lbandini@bu.edu)
Charles Drum (Charles.drum@unh.edu)
Karen Luken (karenluken@gmail.com)
Yolanda Suarez-Balcazar (ysuarez@uic.edu)
Ian D Graham (igraham@ohri.ca)

Version: 4 Date: 21 July 2014

Author's response to reviews: see over
Dear Implementation Science Editorial Team:

Please find the resubmission of manuscript #5178022631236746. We greatly appreciate the time and edits from the editorial team and feel that the manuscript is ready for publication. We have accepted all of the changes that the editor made and addressed minor questions. The revised title provides a much clearer indication of the study’s implications and will draw a larger readership to IS from disability and health promotion researchers (GRAIDs: A framework for closing the gap in the availability of health promotion programs and interventions for people with disabilities.) We have also moved the funding source from the author/title page to its own ‘Funding’ section at the back of the manuscript (before the references). Along with the funding section, we added an acknowledgement to our expert panel members.

We hope it is ok that we added a short paragraph (first paragraph, p. 20) as requested by our funding agency’s federal program officer, Dr. Margaret Campbell. She felt it was a critical piece that needed to be included in this first paper in order to prompt/encourage researchers and practitioners to test the GRAIDs in their respective settings.

Again, thank you for your time and patience in this review process. It is encouraging to know that this may be your first paper in the area of knowledge adaptation and translation for people with disabilities. Our hope is that it will encourage more papers from this community of researchers.

Sincerely,

James H. Rimmer, Ph.D., Professor
Lakeshore Foundation Endowed Chair in Health Promotion and Rehabilitation Sciences
Director of the UAB/Lakeshore Foundation Research Collaborative
Senior Scientist in the UAB Nutrition Obesity Research Center and Center for Exercise Medicine, School of Health Professions, SHPB 331
University of Alabama at Birmingham